Author's response to reviews

Title: Effects of maternal education on diet, anemia, and iron deficiency in Korean school-aged children

Authors:

Hyeon-Jeong Choi (hyunjeong79@hanmail.net)
Hye-Ja Lee (hyejalee@yahoo.co.kr)
Han Byul Jang (hanbyeol-black@hanmail.net)
Ju Yeon Park (jypark5587@gmail.com)
Jae-Heon Kang (fmleader@nuri.net)
Kyung-Hee Park (not@valid.com)
Jihyun Song (jhsong10@korea.kr)

Version: 6 Date: 14 November 2011

Author's response to reviews:

Dear The BMC Series Editorial Production Team,

We checked our manuscript carefully for any small typographical errors and have made table formatting changes.

Sincerely,

Hye-Ja Lee